\documentclass{article}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\setlength{\paperwidth}{24 cm}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{amsmath,amssymb,mathrsfs}
\usepackage{amsthm}
\usepackage[colorlinks]{hyperref}
\usepackage[nameinlink]{cleveref}
\usepackage{url}
\usepackage{graphicx,color,epsfig}
%\usepackage{epstopdf}
\usepackage{amsfonts}
\usepackage{caption}
\usepackage{enumerate}
\usepackage{enumitem}
\usepackage{float}
\usepackage{booktabs}
\usepackage{graphicx}
\usepackage{caption}
\usepackage{subcaption}
\usepackage{array}
\usepackage{booktabs}
\usepackage{fancyhdr}
\usepackage{eurosym}
\usepackage{cite}
\usepackage{upgreek}
\usepackage{enumitem}
\newlist{steps}{enumerate}{1}
\setlist[steps, 1]{label = Step \arabic*:}
\usepackage{multicol}
\hypersetup{pdftitle={},% title
    pdfauthor={Lanre Akinyemi},     % author
    %pdfsubject={Differential Equations},   % subject of the document
    pdfcreator={pdftex},   % creator of the document
    pdfproducer={Texmaker}, % producer of the document
    pdfkeywords={}, % list of keywords
    colorlinks=true,       % false: boxed links; true: colored links
    linkcolor=blue,          % color of internal links
    citecolor=blue,        % color of links to bibliography
    filecolor=magenta,      % color of file links
    urlcolor=blue           % color of external links
}
\setcounter{MaxMatrixCols}{10}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\capu}{\mathcal{D}}
\DeclareMathOperator{\csch}{csch}
\DeclareMathOperator{\sech}{sech}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\renewcommand{\equationautorefname}{Eq.}
\Crefname{figure}{Fig.}{Figs.}
\Crefname{table}{Tab.}{Tabs.}
\makeatletter
\let\oldtheequation\theequation
\renewcommand\tagform@[1]{\maketag@@@{\ignorespaces#1\unskip\@@italiccorr}}
\renewcommand\theequation{(\oldtheequation)}
\makeatother
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newtheorem{theorem}{Theorem}
\newtheorem{case}{Case}
\newtheorem{definition}{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{lemma}{Lemma}
\newtheorem{remark}{Remark}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\topmargin=-2cm \oddsidemargin=0cm \textheight 23.3cm \textwidth 18.5cm
\renewcommand{\baselinestretch}{1.2}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\date{}
\begin{document}
\begin{thebibliography}{10}

\bibitem{al2021downhill}
Bashair Al-Ahmadi, Anwar Saleh, and Wafa Al-Shammakh, \emph{Downhill
zagreb
  polynomials of graphs}, Research \& Reviews: Discrete Mathematical Structures
  \textbf{7} (2021), no.~3, 15--26.

\bibitem{balmaceda2021evaluation}
Rocio Balmaceda-Huarte, Matias~Ezequiel Olmo, Maria~Laura Bettolli,
and
  Maria~Mercedes Poggi, \emph{Evaluation of multiple reanalyses in reproducing
  the spatio-temporal variability of temperature and precipitation indices over
  southern south america}, International Journal of Climatology (2021).

\bibitem{banjare2021studies}
Ganesh~Ram Banjare, DP~Bisen, N~Brahme, and Chitrkant Belodhiya,
\emph{Studies
  on structural properties, luminescence behavior and zeta potential of dy3+
  doped alkaline earth ortho-silicate phosphors}, Materials Science and
  Engineering: B \textbf{263} (2021), 114882.

\bibitem{boyer1997synthesis}
L~Boyer, J~Carpena, and JL~Lacout, \emph{Synthesis of
phosphate-silicate
  apatites at atmospheric pressure}, Solid State Ionics \textbf{95} (1997),
  no.~1-2, 121--129.

\bibitem{costa2021chemical}
Paulo Costa, Joel~S Evangelista, Igor Leal, and Paulo~CML Miranda,
  \emph{Chemical graph theory for property modeling in qsar and qsprcharming
  qsar \& qspr}, Mathematics \textbf{9} (2021), no.~1, 60.

\bibitem{KCdas1}
K.~C. Das, A.~S. Çevik, I.~N. Cangul, and Y.~Shang, \emph{On sombor
index},
  Symmetry \textbf{13} (2021), no.~1, 140.

\bibitem{ding2021degree}
Lei Ding, Syed~Ahtsham Ul~Haq~Bokhary, Masood~Ur Rehman, Usman Ali,
Hirra
  Mubeen, Quaid Iqbal, and Jia-Bao Liu, \emph{Degree-based indices of some
  complex networks}, Journal of Mathematics \textbf{2021} (2021).

\bibitem{fajtlowicz1988conjectures}
Siemion Fajtlowicz, \emph{On conjectures of graffiti}, Annals of
Discrete
  Mathematics, vol.~38, Elsevier, 1988, pp.~113--118.

\bibitem{gutman2021some}
Ivan Gutman, \emph{Some basic properties of sombor indices}, Open
Journal of
  Discrete Applied Mathematics \textbf{4} (2021), no.~1, 1--3.

\bibitem{haddaji2021effect}
Younesse Haddaji, Hicham Majdoubi, Said Mansouri, Youssef Tamraoui,
Rachid
  Boulif, Mina Oumam, and Hassan Hannache, \emph{Effect of sodium
  hexafluorosilicate addition on the properties of metakaolin based geopolymers
  cured at ambient temperature}, Silicon \textbf{13} (2021), 1441--1451.

\bibitem{hester2021quantum}
Gavin~Lee Hester, \emph{Quantum magnetism in the rare-earth
pyrosilicates},
  Ph.D. thesis, Colorado State University, 2021.

\bibitem{irfan2021m}
Muhammad Irfan, Hamood~Ur Rehman, Hassan Almusawa, Saffina Rasheed,
and
  Imran~Abbas Baloch, \emph{M-polynomials and topological indices for line
  graphs of chain silicate network and h-naphtalenic nanotubes}, Journal of
  Mathematics \textbf{2021} (2021).

\bibitem{kullib}
VR~Kulli, \emph{The (a, b)-ka temperature indices of tetrameric 1,
  3-adamantane}.

\bibitem{liebau2012structural}
Friedrich Liebau, \emph{Structural chemistry of silicates:
structure, bonding,
  and classification}, Springer Science \& Business Media, 2012.

\bibitem{liu2021graph}
Jia-Bao Liu, Muhammad Imran, Shakila Baby, Hafiz Muhammad~Afzal
Siddiqui, and  Muhammad~Kashif Shafiq, \emph{Graph indices for cartesian product of f-sum of
  connected graphs.}, Combinatorial Chemistry \& High Throughput Screening
  (2021).

\bibitem{mandlimath2021synthesis}
Triveni~Rajashekhar Mandlimath, Daneshwaran Balaji, and
Sathasivam~Pratheep
  Kumar, \emph{Synthesis, structural and thermal expansion investigation of la,
  ce and eu substituted bi4 (sio4) 3}, Materials Chemistry and Physics (2021),
  124841.

\bibitem{mondal2021qspr}
Sourav Mondal, Arindam Dey, Nilanjan De, and Anita Pal, \emph{Qspr
analysis of
  some novel neighbourhood degree-based topological descriptors}, Complex \&
  Intelligent Systems \textbf{7} (2021), no.~2, 977--996.

\bibitem{monsalve2021vertex}
Juan Monsalve and Juan Rada, \emph{Vertex-degree based topological
indices of digraphs}, Discrete Applied Mathematics \textbf{295} (2021), 13--24.

\bibitem{narayankarharmonic}
Kishori~P Narayankar, Afework~Teka Kahsay, and Dickson Selvan,
\emph{Harmonic temperature index of certain nanostructures}.

\bibitem{selvarani2021generalization}
P~Selvarani, K~Dhanalakshm, and J~Irudaya~Monica Catherine,
  \emph{Generalization of new degree based topological indices of silicate
  network graph}, Journal of Physics: Conference Series, vol. 1724, IOP
  Publishing, 2021, p.~012034.

\bibitem{javaid2017m}
Javaid, Muhammad, and Chahn Yong Jung. \emph{ M-Polynomials and topological indices of silicate and oxide networks.} International Journal of Pure and Applied Mathematics 115, no. 1 (2017): 129-152.
\bibitem{javaid2017topological}
Javaid, Muhammad, Masood Ur Rehman, and Jinde Cao. \emph{Topological indices of rhombus type silicate and oxide networks.} Canadian Journal of Chemistry 95, no. 2 (2017): 134-143.
\bibitem{sigarreta2021mathematical}
Jos{\'e}~M Sigarreta, \emph{Mathematical properties of variable topological   indices}, Symmetry \textbf{13} (2021), no.~1, 43.

\bibitem{soares2021new}
Viviane~O Soares, Francisco~C Serbena, Ivan Mathias, Murilo~C
Crovace, and
  Edgar~D Zanotto, \emph{New, tough and strong lithium metasilicate dental
  glass-ceramic}, Ceramics International \textbf{47} (2021), no.~2, 2793--2801.

\bibitem{wei2020some}
Chang-Cheng Wei, Muhammad Salman, Usman Ali, Masood~Ur Rehman,
Muhammad~Aqeel
  Ahmad~Khan, Muhammad~Hasanain Chaudary, and Farooq Ahmad, \emph{Some
  topological invariants of graphs associated with the group of symmetries},
  Journal of Chemistry \textbf{2020} (2020).

\bibitem{zakharov2021topological}
Anton~B Zakharov, Dmytro~K Tsarenko, and Vladimir~V Ivanov,
\emph{Topological
  characteristics of iterated line graphs in the qsar problem: a multigraph in
  the description of properties of unsaturated hydrocarbons}, Structural
  Chemistry (2021), 1--11.

\end{thebibliography}

\title{Characterization of temperature indices of silicates}
\author{Abdul Rauf Khan$^{1},\,$ Muhammad Usman Ghani$^{2},\,$ Abdul Ghaffar$^{1},\,$  \\Hafiz Muhammad Asif$^{3},\,$ Mustafa Inc$^{4,5}$\\
\\
\\
$^{1}$\textit{Department of Mathematics, Faculty of Science, Ghazi University 32200, Dera Ghazi Khan, Punjab, Pakistan.}\\
$^{2}$\textit{Department of Mathematics, Khawaja Fareed University of Engineering, Information Technology, 64200, Rahim Yar Khan, Punjab, Pakistan}\\
$^{3}$\textit{DInstitute of Chemical Sciences, Baha Uddin Zakariya University, Multan, Punjab, Pakistan.}\\
$^{4}$\textit{Firat University, Science Faculty, Department of Mathematics, 23119
Elazig, Turkey}\\
$^{5}$\textit{Department of Medical Research, China Medical University, 40402 Taichung, Taiwan}\\
\\
\\
\textit{e-mail: khankts@gmail.com, abdulghaffar.jaffar@gmail.com,\,musmanghani85a@gmail.com,}\\
\textit{hmasifc@bzu.edu.pk,\,minc@firat.edu.tr}}

\maketitle
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
    %%%%%% Abstract %%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{abstract}
\noindent
A topological index is a numerical value that describes the entire structure of a chemical compound's molecular graph and aids in the understanding of its physical properties, chemical reactivities, and boiling activities. In chemical graph theory, these indices are instrumental in quantifying various chemical properties of chemical compounds such as $SiO_{4}$. The selection of $SiO_{4}$ in this work is an important aspect because of its structural flexibility, easy availability, and extraordinary nature to find its numerical values. In this paper, we compute systematically the first and second temperature, hyper temperature indices, the sum connectivity temperature index, the product connectivity temperature index, the reciprocal product connectivity temperature index, and the F temperature index of a molecular graph $SiO_{4}$ embedded in a chain. All of the discussed interlinked networks in this manuscript are well motivated by the molecular structure of $SiO_{4}$.
\noindent
\textbf{Keywords:} Temperature indices, $SiO_{4}$ embedded in a chain.
%{\color{red}}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
   %%%% Introduction %%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\end{abstract}
\section{Introduction}
Using chemical graph theory, one can determine a wide range of characteristics such as chemical networks, physical, chemical, and thermal properties, biological activity, and chemical activity \cite{gutman2021some}. Topological indices, which are molecular descriptors, can characterize these features and specific graphs \cite{sigarreta2021mathematical}. In chemical graph theory, vertices represent atoms in a chemical compound whereas edges indicate chemical bonding between the atoms \cite{wei2020some}. The topological index of a chemical composition is a numerical value or a continuation of a given structure under discussion, which indicates chemical, physical and biological properties of a structure of chemical molecule \cite{KCdas1,ding2021degree,irfan2021m}.\\
Mathematical chemistry explains the way to use polynomials and functions to provide instructions hidden in the symmetry of molecular graphs, and graph theory has many applications in modern chemistry, particularly organic chemistry. Many applications of topological indices are used in theoretical chemistry, particularly qualitative structure property relationships (QSPR)/ qualitative structure activity relationships (QSAR) research. Many well-known researchers have investigated topological indices in order to learn more about various graph families \cite{costa2021chemical,mondal2021qspr}. For example, in QSPR and QSAR, topological indices are employed as handy numerical descriptors in comparison with biological, physical and chemical parameters of molecules, which is an advantage for chemical industry. Many researchers have worked on various chemical compounds and computed topological descriptors of various molecular graphs during recent years \cite{al2021downhill,javaid2017m,javaid2017topological,zakharov2021topological}.\\
Graph indices have found a number of applications in a wide range of fields in chemistry including chemical documentation, structure-property relationships, structure-activity relationships, and pharmaceutical drug design. The general calculation of graph indices has attracted a lot of interest in mathematics \cite{liu2021graph}. \\
We only consider finite, simple, connected graphs in this paper. Assume that $G$ is a graph with the vertex set $V_G$ and the edge set $E_G$. The number of vertices adjacent to a vertex $u$ determines its degree $d_u$. For fundamental notations and terminologies, we refer the reader to \cite{monsalve2021vertex}.\\
Fajtlowicz put forward the follow definition of the temperature of any vertex $u$ of a graph $G$ \cite{fajtlowicz1988conjectures}:
\begin{equation}\label{T}
  T_{u_i} = \frac{d_{u_i}}{|V_G|-d_{u_i}} \,\,\qquad where \,\,\qquad \forall~~u_{i}\in V_G
\end{equation}
The first temperature index \cite{narayankarharmonic} is introduced as follows:
\begin{equation}\label{T1}
  T_{1}(G) = \sum_{uv\in E(G)} \left(T_u +T_v\right)
\end{equation}
In 2020, Kulli introduced second temperature index \cite{kullib}, which is given by
\begin{equation}\label{T2}
  T_{2}(G) = \sum_{uv\in E(G)} \left(T_u \times T_v\right)
\end{equation}
Kulli investigated the first and second hyper temperature indices in \cite{kullib}, which are defined as
\begin{equation}\label{HT1}
 HT_{1}(G) = \sum_{uv\in E(G)} \left(T_u +T_v\right)^{2}
\end{equation}
\begin{equation}\label{HT2}
  HT_{2}(G) = \sum_{uv\in E(G)} \left(T_u \times T_v\right)^{2}
\end{equation}
In the same paper \cite{kullib}, some other related topological indices are introduced. The sum connectivity temperature index, the product connectivity temperature index, and the reciprocal product connectivity index are defined respectively as
\begin{equation}\label{ST}
 ST(G) = \sum_{uv\in E(G)}\frac{1}{ \sqrt{\left(T_u +T_v\right)}}
\end{equation}
\begin{equation}\label{PT}
 PT(G) = \sum_{uv\in E(G)}\frac{1}{ \sqrt{\left(T_{u} \times T_{v}\right)}}
\end{equation}
\begin{equation}\label{RPT}
 RPT(G) = \sum_{uv\in E(G)}{ \sqrt{\left(T_u \times T_v\right)}}
\end{equation}
Kulli also studied the F-temperature index and general temperature index of a graph $G$ in \cite{kullib}, and they are defined as\\
\begin{equation}\label{FT}
 FT(G) = \sum_{uv\in E(G)}\left(T_{u}^{2} +T_{v}^{2}\right)
\end{equation}
In industrial chemistry, a silicate $Si$ is an element of a family of anions (an ion is a atom or molecule with a net-electrical charge) containing of silicon and oxygen, Boyer used the general formula $\left[SiO{_{4-t}^{(4-2t)-}}\right]_{n}$ for $0\leq t<2$, in \cite{boyer1997synthesis}. Some researchers also explain the family of anions by using this formula, the Orthosilicate family $SiO_{4}^{4-}(t=0)$, see in \cite{banjare2021studies}, metasilicate $SiO_{3}^{2-}(t=1)$, see in \cite{soares2021new} and pyrosilicate $Si_{2}O_{7}^{6-}(t=\frac{1}{2}, n=2)$, see in \cite{hester2021quantum}. We can extend silicate $Si$ to any anions containing silicon (atom-bonding with other than oxygen atoms), as hexafluorosilicate $SiF_{3}^{2-}$, see in \cite{haddaji2021effect}. Here, we discuss only chain of silicates, which is obtained by alternating sequence of tetrahedron $SiO_{4}$ \cite{mandlimath2021synthesis,selvarani2021generalization}. \\
In this article, the above  defined eight temperature indices are constructed by the atom-bonds partition of chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$, which are partitioned according to the degrees of its $Si$ and oxygen atoms. We also investigate silicon tetrahedron $Si O_{4}$ in a compound structure and derive the precise formulas of certain essential degree-based temperate indices using the approach of atom-bonds partitioning of molecular structure of silicates. We obtain the the concept of temperature indices from \cite{balmaceda2021evaluation,kullib}.

\section{Chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$}
The basic unit of silicates is empirically represented by formula $SiO_{4}$, possessing tetrahedron geometry \cite{liebau2012structural}. Almost all the silicates contain $SiO_{4}$ tetrahedron. From chemical point of view, a tetrahedron $SiO_{4}$, as shown in Figure \ref{SC}, containing oxygen atoms at the four corners of tetrahedron, and the silicon atom is bonded with equally spaced atoms of oxygen. From resulting $SiO_{4}$, a silicate tetrahedron, joins with other $SiO_{4}$ horizontally, a single chain of silicates is obtained. Similarly, when two molecules of $SiO_{4}$ join corner to corner, then each $SiO_{4}$ shares its oxygen atoms with the other $SiO_{4}$ molecule, as in Figure \ref{SC}. After completing this process of sharing, these two molecules of $SiO_{4}$, can be joined with two other molecules. Now, we get a chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$, where $p$ and $q$ are the numbers of silicate chains formed and total number of $SiO_4$ in one silicate chain, respectively. Here, in chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$, $pq$ number of tetrahedron $SiO_{4}$ is used; see Figure \ref{SC}.
\begin{figure}[H]
  \centering
\includegraphics[width=13cm]{cs.eps}
   \caption{Schematic illustration of chain of silicates: where circular blue and black elliptical shapes showing the oxygen the silicon atoms, respectively}\label{SC}
\end{figure}
Here, in chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$, we observed that there are three type of atom-bonds on the bases of valency of every atom of $\mathcal{S}\mathcal{C}_{q}^{p}$. Therefore, there are two types of atoms $v_i$ and $v_j$, such that $d_{v_i}=3$ and $d_{v_j}=6$, where $d_{v_i}$ and $d_{v_j}$ means the valency of atoms $\forall $ $v_i,v_{j}\in \mathcal{S}\mathcal{C}_{q}^{p}$. According to the valencies (3 and 6) of atoms, there are three types of atom-bonds, which are $(3\sim3)$, $(3\sim6)$ and $(6\sim6)$ in $\mathcal{S}\mathcal{C}_{q}^{p}$. On the base of valency, Table \ref{tab1} provides the partition of the set of atom-bonds. The atom-bonds partition of $\mathcal{S}\mathcal{C}_{q}^{p}$ shown as:
\begin{eqnarray*}
   %\nonumber % Remove numbering (before each equation)
E_{(3,3)} &=& \Big{\{}e=u\sim v,\forall ~u,v\in V(\mathcal{S}\mathcal{C}_{q}^{p} )\Big{|} d_{u}=3,d_{v}=3\Big{\}}, |E_{(3,3)}|=3p+2\\
     E_{(3,6)} &=& \Big{\{}e=u\sim v,\forall ~u,v\in V(\mathcal{S}\mathcal{C}_{q}^{p} )\Big{|} d_{u}=3,d_{v}=6\Big{\}},~|E_{(3,6)}|=3(pq+q)-4\\
      E_{(6,6)} &=& \Big{\{}e=u\sim v,\forall ~u,v\in V(\mathcal{S}\mathcal{C}_{q}^{p} )\Big{|} d_{u}=6,d_{v}=6\Big{\}},~|E_{(6,6)}|=3(pq-2q)+2.
  \end{eqnarray*}
  The total number of atoms and atom-bonds $\mathcal{S}\mathcal{C}_{q}^{p}$:
\begin{equation*}\label{0}
  |V(\mathcal{S}\mathcal{C}_q^p)| = 3p^2-p  \qquad\qquad and \qquad \qquad |E(\mathcal{S}\mathcal{C}_q^p)| = 6pq+3p-q
\end{equation*}
By using equation (\ref{T}) and above partition of $\mathcal{S}\mathcal{C}_{q}^{p}$, three types of edges on the basis of the temperature of end vertices of an edge can be identified. They are described in Table \ref{tab1}.
\begin{table}[H]
\center
\caption{Atomic-bond partition of $\mathcal{S}\mathcal{C}_{q}^{p}$, for $p = q$}\label{tab1}
\begin{tabular} {|c|c|c| c|}
 \hline
 $(T_u , T_v)$&$\left(\frac{3}{(3p^2-p)-3},\frac{3}{(3p^2-p)-3}\right)$  &$\left(\frac{3}{(3p^2-p)-3},\frac{6}{(3p^2-p)-6}\right)$  &$\left(\frac{6}{(3p^2-p)-6},\frac{6}{(3p^2-p)-6}\right)$   \\
 \hline
Frequency&  $3p+2$&$3(pq+q)-4$&$3(pq-2q)+2$\\
  \hline
\end{tabular}
\end{table}

\subsection{Temperature indices for chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$ $p = q$}


\begin{theorem}\label{thm1}
Let $\mathcal{S}\mathcal{C}_{q}^{p}$ be a chain of silicates. Then the first temperature index is $\frac{6(3p+2)}{3p^2-p-3}+\frac{9(3p^2+3p-4)(3p^2-p-4)}{(3p^2-p-3)(3p^2-p-6)}+\frac{12(3p^2-6p+2)}{3p^2-p-6}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the first temperature index (\ref{T1}), we obtain
\begin{eqnarray*}
T_{1}(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\left(T_3 +T_3 \right)+ \sum_{E_{(3, 6)}} \left(T_3 +T_6\right) + \sum_{E_{(6, 6)}} \left(T_6 +T_6\right)\\
&=&(3p+2)\left[\frac{3}{(3p^2-p)-3}+\frac{3}{(3p^2-p)-3} \right]\\
&+&(3p^2+3p-4)\left[\frac{3}{(3p^2-p)-3}+\frac{6}{(3p^2-p)-6} \right]\\
&+&(3p^2-6p+2)\left[\frac{6}{(3p^2-p)-6}+\frac{6}{(3p^2-p)-6} \right]
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{t1}
T_1(\mathcal{S}\mathcal{C}_{q}^{p})=&\frac{6(3p+2)}{3p^2-p-3}+\frac{9(3p^2+3p-4)(3p^2-p-4)}{(3p^2-p-3)(3p^2-p-6)}+\frac{12(3p^2-6p+2)}{3p^2-p-6}.
\end{eqnarray}
\end{proof}
\begin{theorem}\label{thm2}
Let $\mathcal{S}\mathcal{C}_{q}^{p}
$ be a chain of silicates. Then the second temperature index is $\frac{9(3p+2)}{(3p^2-p-3)^2}+\frac{18(3p^2+3p-4)(3p^2-p-4)}{(3p^2-p-3)(3p^2-p-6)}+\frac{36(3p^2-6p+2)}{(3p^2-p-6)^2}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the second temperature index (\ref{T2}), we obtain
\begin{eqnarray*}
T_{2}(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\left(T_3 \times T_3 \right)+ \sum_{E_{(3, 6)}} \left(T_3 \times T_6\right) + \sum_{E_{(6, 6)}} \left(T_6 \times T_6\right)\\
&=&(3p+2)\left[\frac{3}{(3p^2-p)-3}\times \frac{3}{(3p^2-p)-3} \right]\\
&+&(3p^2+3p-4)\left[\frac{3}{(3p^2-p)
-3}\times \frac{6}{(3p^2-p)-6} \right]\\
&+&(3p^2-6p+2)\left[\frac{6}{(3p^2-p)-6}\times \frac{6}{(3p^2-p)-6} \right]
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{t2}
T_2(\mathcal{S}\mathcal{C}_{q}^{p})=\frac{9(3p+2)}{(3p^2-p-3)^2}+\frac{18(3p^2+3p-4)}{(3p^2-p-3)(3p^2-p-6)}+\frac{36(3p^2-6p+2)}{(3p^2-p-6)^2}.
\end{eqnarray}
\end{proof}
\begin{theorem}\label{thm3}
Let $\mathcal{S}\mathcal{C}_{q}^{p}
$ be a chain of silicates. Then the first hyper temperature index is $\frac{36(3p+2)}{(3p^2-p-3)^2}+\frac{81(3p^2+3p-4)(3p^2-p-4)^2}{(3p^2-p-3)^2 (3p^2-p-6)^2}+\frac{144(3p^2-6p+2)}{(3p^2-p-6)^2}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the first hyper temperature index (\ref{HT1}), we obtain
\begin{eqnarray*}
HT_{1}(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\left(T_3 +T_3 \right)^{2}+ \sum_{E_{(3, 6)}} \left(T_3 +T_6\right)^{2} + \sum_{E_{(6, 6)}} \left(T_6 +T_6\right)^{2}\\
&=&(3p+2)\left[\frac{3}{(3p^2-p)-3}+\frac{3}{(3p^2-p)-3} \right]^{2}\\
&+&(3p^2+3p-4)\left[\frac{3}{(3p^2-p)
-3}+\frac{6}{(3p^2-p)-6} \right]^{2}\\
&+&(3p^2-6p+2)\left[\frac{6}{(3p^2-p)
-6}+\frac{6}{(3p^2-p)-6} \right]^{2}
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{Ht1}
HT_1(\mathcal{S}\mathcal{C}_{q}^{p})=\frac{36(3p+2)}{(3p^2-p-3)^2}+\frac{81(3p^2+3p-4)(3p^2-p-4)^2}{(3p^2-p-3)^2 (3p^2-p-6)^2}+\frac{144(3p^2-6p+2)}{(3p^2-p-6)^2}.
\end{eqnarray}
\end{proof}
\begin{theorem}\label{thm4}
Let $\mathcal{S}\mathcal{C}_{q}^{p}$ be a chain of silicates. Then the second hyper temperature index is $\frac{81(3p+2)}{(3p^2-p-3)^4}+\frac{324(3p^2+3p-4)}{(3p^2-p-3)^2 (3p^2-p-6)^2}+\frac{1296(3p^2-6p+2)}{(3p^2-p-6)^4}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the second temperature index (\ref{HT2}), we obtain
\begin{eqnarray*}
HT_{2}(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\left(T_3 \times T_3 \right)^{2}+ \sum_{E_{(3, 6)}} \left(T_3 \times T_6\right)^{2} + \sum_{E_{(6, 6)}} \left(T_6 \times T_6\right)^{2}\\
&=&(3p+2)\left[\frac{3}{(3p^2-p)-3}\times \frac{3}{(3p^2-p)-3} \right]^{2}\\
&+&(3p^2+3p-4)\left[\frac{3}{(3p^2-p)-3}\times \frac{6}{(3p^2-p)-6} \right]^{2}\\
&+&(3p^2-6p+2)\left[\frac{6}{(3p^2-p)
-6}\times \frac{6}{(3p^2-p)-6} \right]^{2}
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{Ht2}
HT_2(\mathcal{S}\mathcal{C}_{q}^{p}
)=\frac{81(3p+2)}{(3p^2-p-3)^4}+\frac{324(3p^2+3p-4)}{(3p^2-p-3)^2 (3p^2-p-6)^2}+\frac{1296(3p^2-6p+2)}{(3p^2-p-6)^4}.
\end{eqnarray}
\end{proof}
\begin{theorem}\label{thm5}
Let $\mathcal{S}\mathcal{C}_{q}^{p}$ be a chain of silicates. Then the sum connectivity temperature index is $\frac{(3p+2)\sqrt{(3p^2-p-3)}}{\sqrt{6}}+\frac{(3p^2+3p-4)\sqrt{(3p^2-p-3) (3p^2-p-6)}}{3\sqrt{(3p^2-p-4)}}+\frac{\sqrt{(3p^2-p-6)}}{2(3p^2-6p+2)\sqrt{3}}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the sum connectivity temperature index (\ref{ST}), we obtain
\begin{eqnarray*}
ST(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\frac{1}{\sqrt{\left(T_3 + T_3 \right)}}+ \sum_{E_{(3, 6)}} \frac{1}{\sqrt{\left(T_3 + T_6\right)}} + \sum_{E_{(6, 6)}} \frac{1}{\sqrt{\left(T_6 + T_6\right)}}\\
&=&(3p+2)\frac{1}{\sqrt{\left[\frac{3}{(3p^2-p)
-3}+ \frac{3}{(3p^2-p)-3} \right]}}\\
&+&(3p^2+3p-4)\frac{1}{\sqrt{\left[\frac{3}{(3p^2-p)-3}+ \frac{6}{(3p^2-p)-6} \right]}}\\
&+&(3p^2-6p+2)
\frac{1}{\sqrt{\left[\frac{6}{(3p^2-p)-6}+ \frac{6}{(3p^2-p)-6} \right]}}
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{St}
ST(\mathcal{S}\mathcal{C}_{q}^{p}
)=&\frac{(3p+2)\sqrt{(3p^2-p-3)}}{\sqrt{6}}+\frac{(3p^2+3p-4)\sqrt{(3p^2-p-3) (3p^2-p-6)}}{3\sqrt{(3p^2-p-4)}}\nonumber\\
&+\frac{\sqrt{(3p^2-p-6)}}{2(3p^2-6p+2)\sqrt{3}}.
\end{eqnarray}
\end{proof}
\begin{theorem}\label{thm6}
Let $\mathcal{S}\mathcal{C}_{q}^{p}$ be a chain of silicates. Then the product connectivity temperature index is $\frac{9p^3+3p^2-11p-6}{3}+\frac{(3p^2+3p-4)\sqrt{(3p^2-p-3)(3p^2-p-6)}}{3\sqrt{2}}+\frac{(3p^2-6p+2)(3p^2-p-6)}{6}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the product connectivity temperature index (\ref{PT}), we obtain
\begin{eqnarray*}
PT(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\frac{1}{\sqrt{\left(T_3 \times T_3 \right)}}+ \sum_{E_{(3, 6)}} \frac{1}{\sqrt{\left(T_3 \times T_6\right)}} + \sum_{E_{(6, 6)}} \frac{1}{\sqrt{\left(T_6 \times T_6\right)}}\\
&=&(3p+2)\frac{1}{\sqrt{\left[\frac{3}{(3p^2-p)-3}\times \frac{3}{(3p^2-p)-3} \right]}}\\
&+&(3p^2+3p-4)\frac{1}{\sqrt{\left[\frac{3}{(3p^2-p)-3}\times \frac{6}{(3p^2-p)-6} \right]}}\\
&+&(3p^2-6p+2)\frac{1}{\sqrt{\left[\frac{6}{(3p^2-p)-6}\times \frac{6}{(3p^2-p)
-6} \right]}}
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{Pt}\nonumber
PT(\mathcal{S}\mathcal{C}_{q}^{p})&=&\frac{9p^3+3p^2-11p-6}{3}+\frac{(3p^2+3p-4)\sqrt{(3p^2-p-3)(3p^2-p-6)}}{3\sqrt{2}}\\
&+&\frac{(3p^2-6p+2)(3p^2-p-6)}{6}.
\end{eqnarray}
\end{proof}
\begin{theorem}\label{thm7}
Let $\mathcal{S}\mathcal{C}_{q}^{p}$ be a chain of silicates. Then the reciprocal product temperature index is $\frac{18(3p+2)}{(3p^2-p-3)^2}+\frac{(3p^2+3p-4)(45p^4-60p^3-67p^2+48p+72)}{(3p^2-p-3)^2(3p^2-p-6)^2}+\frac{144(3p^2-6p+2)}{(3p^2-p-6)^2}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the second temperature index (\ref{RPT}), we obtain
\begin{eqnarray*}
RPT(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\sqrt{\left(T_3 \times T_3 \right)}+ \sum_{E_{(3, 6)}} \sqrt{\left(T_3 \times T_6\right)} + \sum_{E_{(6, 6)}} \sqrt{\left(T_6 \times T_6\right)}\\
&=&(3p+2)\sqrt{\left[\frac{3}{(3p^2-p)-3}\times \frac{3}{(3p^2-p)-3} \right]}\\
&+&(3p^2+3p-4)\sqrt{\left[\frac{3}{(3p^2-p)
-3}\times \frac{6}{(3p^2-p)-6} \right]}\\
&+&(3p^2-6p+2)\sqrt{\left[\frac{6}{(3p^2-p)-6}\times \frac{6}{(3p^2-p)-6} \right]}
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{t2}
RPT(\mathcal{S}\mathcal{C}_{q}^{p})=\frac{3(3p+2)}{(3p^2-p-3)}+\frac{3(3p^2+3p-4)\sqrt{2}}{\sqrt{(3p^2-p-3)(3p^2-p-6)}}+\frac{6(3p^2-6p+2)}{(3p^2-p-6)}.
\end{eqnarray}
\end{proof}
\begin{theorem}\label{thm8}
Let $\mathcal{S}\mathcal{C}_{q}^{p}$ be a chain of silicates. Then the F-temperature index is $\frac{18(3p+2)}{(3p^2-p-3)^2}+\frac{(3p^2+3p-4)(45p^4-60p^3-67p^2+48p+72)}{(3p^2-p-3)^2(3p^2-p-6)^2}+\frac{144(3p^2-6p+2)}{(3p^2-p-6)^2}$.
\end{theorem}
\begin{proof}
Using the atomic bonds partition from Table \ref{tab1} in the formula of the F-temperature index (\ref{FT}), we obtain
\begin{eqnarray*}
FT(\mathcal{S}\mathcal{C}_{q}^{p}) &=&  \sum_{E_{(3, 3)}}\left(T_{3}^{2} +T_{3}^{2} \right)+ \sum_{E_{(3, 6)}} \left(T_{3}^{2} +T_{6}^{2}\right) + \sum_{E_{(6, 6)}} \left(T_{6}^{2} +T_{6}^{2}\right)\\
&=&(3p+2)\left[\Big{\{}\frac{3}{(3p^2-p)-3}\Big{\}}^{2}+\Big{\{}\frac{3}{(3p^2-p)-3} \Big{\}}^{2}\right]\\
&+&(3p^2+3p-4)\left[\Big{\{}\frac{3}{(3p^2-p)-3}\Big{\}}^{2}+\Big{\{}\frac{6}{(3p^2-p)-6} \Big{\}}^{2}\right]\\
&+&(3p^2-6p+2)
\left[\Big{\{}\frac{6}{(3p^2-p)
-6}\Big{\}}^{2}+\Big{\{}\frac{6}{(3p^2-p)-6}\Big{\}}^{2} \right]
\end{eqnarray*}
After simplification, we get
\begin{eqnarray}\label{Ft}
FT(\mathcal{S}\mathcal{C}_{q}^{p})=&\frac{18(3p+2)}{(3p^2-p-3)^2}+\frac{(3p^2+3p-4)(45p^4-60p^3-67p^2+48p+72)}{(3p^2-p-3)^2(3p^2-p-6)^2}\nonumber\\
&+\frac{144(3p^2-6p+2)}{(3p^2-p-6)^2}.
\end{eqnarray}
\end{proof}

\subsection{Numerical and graphical comparison of temperature indices for chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$}
In this section, we perform a numerical and graphical comparison temperature indices for $n = 2,3,4,...,12$ of chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$.
\begin{table}[H]
\centering
\caption{Temperature indices of chain of silicates $\mathcal{S}\mathcal{C}_{q}^{p}$ for $p=q$}\label{tab3}
\begin{tabular} {|c| c| c| c| c| c| c| c| c|}
        \hline
         $p$&$T_1$ & $T_2$ & $HT_1$& $HT_2$& $ST$ &$PT$ &$ RPT$&$FT$ \\
    \hline
2&  36.86&  14.970& 75.95&  654.06& 19.01&  37.47&      17.65&  23.44\\
3&  22.05&  2.970&  13.04&  792.08& 67.064& 256.64&     12.22&  5.71\\
4&  19.9&   1.370&  5.88&   886.76& 153.16& 876.99&     10.73&  2.9\\
5&  17.45&  0.808&  3.439&  951.75& 187.92& 2208.36&    11.14&  1.81\\
6&  17.17&  0.538&  2.281&  999  &  481.76& 4647.35&    10.54&  1.25\\
7&  16.73&  0.387&  1.63&   1034.86&745.07& 8677.45&    10.42&  0.92\\
8&  16.44&  0.292&  1.23&   1063.01&1088.26&14869.11&   10.36&  0.71\\
9&  16.24&  0.228&  0.96&   1085.68&1521.71&23879.61&   10.32&  0.56\\
10& 16.08&  0.180&  0.77&   1104.34&2055.82&36453.22&   10.29&  0.46\\
11& 15.96&  0.150&  0.63&   1085.68&2700.99&53421.06&   10.27&  0.38\\
12& 15.86&  0.120&  0.53&   1104.34&3467.591&7501.22&   10.26&  0.32\\

        \hline
\end{tabular}
\end{table}
\begin{figure}[H]
  \centering
\includegraphics[width=12cm]{TTTT.eps}
   \caption{2D-comparison of temperature indices of $\mathcal{S}\mathcal{C}_{q}^{p}$}\label{2D}
\end{figure}
\begin{figure}[H]
  \centering
\includegraphics[width=12cm]{3DT.eps}
   \caption{3D-comparison of temperature indices of $\mathcal{S}\mathcal{C}_{q}^{p}$}\label{3D}
\end{figure}
\section{Conclusion}
In this paper, we compute the first temperature index, second temperature index, first hyper temperature index, second hyper temperature index, sum temperature index, product temperature, reciprocal product temperature index, and F-temperature index of silicate network and silicate chain network, which correlates well with entropy, acentric factor, enthalpy of vaporization, and standard enthalpy of vaporization. In QSPR/QSAR research, topological indices such as the Zagreb index, Randic index, and atom-bond connectivity index are utilized to predict chemical compound bioactivity. This study disclosed new avenues in the field of topological index for silicates networks.
\section{Open problems}
It is natural to arise a question about the characterization of chain of silicates on the basis of the nature of temperature indices. For the characterization of the chain of silicates, the following open problems seem to be interesting.
\begin{itemize}
  \item Is the temperature indices effected, when both $p$ and $q$ are even or odd, one is even other is odd, $p<q$?
  \item What about the case $p\geq q$?
\end{itemize}
\noindent
{\bf Funding} Not Applicable.\\
{\bf Author Contributions} [Abdul Rauf Khan], [Muhammad Usman Ghani] and [Abdul Ghaffar] conceived the theoretical model. [Abdul Rauf Khan], [Hafiz Muhammad Asif] and [Mustafa Inc] performed the calculations and plotted the graphs. They analyzed and discussed the results with [Muhammad Usman Ghani] and [Hafiz Muhammad Asif].\\
{\bf Data Availability} All data generated or analysed during this study are included in this article.\\\\
{\bf Declarations}\\
{\bf Competing Interests} The authors declare no competing interests.\\
{\bf Ethics Approval} Not Applicable.\\
{\bf Consent to Participate} Not Applicable.\\
{\bf Consent for Publication} The authors declare that the figures and tables used in this paper are original and are not published anywhere.\\
{\bf Conflict of Interest} The authors declare that they have no conflict of interest


%%%-----------------------------------------------------------------------
%\bibliographystyle{amsplain}
%\bibliography{references}
\providecommand{\bysame}{\leavevmode\hbox
to3em{\hrulefill}\thinspace}
\providecommand{\MR}{\relax\ifhmode\unskip\space\fi MR }
% \MRhref is called by the amsart/book/proc definition of \MR.
\providecommand{\MRhref}[2]{%
  \href{http://www.ams.org/mathscinet-getitem?mr=#1}{#2}
} \providecommand{\href}[2]{#2}



\end{document}
